Author's response to reviews

**Title:** Reproducibility of wrist home blood pressure measurement with position sensor and automatic data storage

**Authors:**

- Sakir Uen (suen@uni-bonn.de)
- Rolf Fimmers (rolf.fimmers@ukb.uni-bonn.de)
- Miriam Brieger (miriam.brieger@gmx.de)
- Georg Nickenig (georg.nickenig@ukb.uni-bonn.de)
- Thomas Mengden (mengden@uni-bonn.de)

**Version:** 5  **Date:** 29 January 2009

**Author's response to reviews:**

**Major revisions:**

1) Author list - The list of authors in the manuscript was written exactly as they are in the submission system.

2) Dr. Sakir Uen is indicated as corresponding author in the title page

3) Authors affiliation was corrected to the right format

4) In the authors contributions section the following sentence "All authors read and approved the final manuscript.

5) Vertical elements are deleted in the tables.

6) There are 3 figures, but figure 1 is subdivided to 1a and 1b, with only one legend for both part of the figure 1, and the third figure is figure 2. If its possible please take this format.

7) The appearance of the references of references 9 and 8 was changed in the text in the order that they appear in the list of references.

**Minor revisions:**

- Email addresses for all the authors were included on the title page.
- Abstract as a heading above the abstract was added.

**Introduction** - renamed as Background.

**Study design** - renamed as Methods.

**Conclusions** - We require a Conclusions section after the Discussion. We suggest you create this from the Limitations of the investigation and the Perspectives sections.
Competing Interests changed to Competing interests.
Acknowledgements section was deleted.
Figure titles and legends - renamed to Figure legends.
A consistent format for the legends was used.